Author's response to reviews

Title: Analysis of EGFR, HER2, and TOP2A gene status and chromosomal polysomy in gastric adenocarcinoma from Chinese patients

Authors:

Zhiyong Liang (liangzhiyong1220@yahoo.com)
Xuan Zeng (zengx88@yahoo.com.cn)
Jie Gao (zy_795@hotmail.com)
Shafei Wu (Suphie@sina.com)
Peng Wang (chinaroc@tom.com)
Xiaohua Shi (shixiaohua3762@hotmail.com)
Jing Zhang (jingzhangsmile@163.com)
Tonghua Liu (liuth_pumch@yahoo.com.cn)

Version: 7 Date: 5 December 2008

Author's response to reviews: see over
Dear editors,
I have changed the formats of our manuscript according to your suggestions.

Manuscript sections include (in the following order): Abstract; Background; Methods; Results; Discussion; Conclusions; Competing interests; Authors’ contributions; Acknowledgements; References; Figure legends; Tables.
I have inserted a Conclusions section after the Discussion.
The reference list contains all authors names.

Best regards,

Tonghua Liu